Author's response to reviews

Title: Characterization of extensively drug-resistant isolates Mycobacterium tuberculosis in Siberia.

Authors:

Maya A. Dymova (maya.a.rot@gmail.com)
Andrey G. Cherednichenko (bact.nniit@gmail.com)
Olga I. Alkhovik (olgaialkhovik@mail.ru)
Eugeniya A. Khrapov (Khrap_off@ngs.ru)
Tatjana I. Petrenko (tipetrenko@gmail.com)
Maxim L. Filipenko (max@nibochnsk.ru)

Version: 2 Date: 23 October 2013

Author's response to reviews:

Dear Editor,

I would like to submit the attached manuscript, “Characterization of extensively drug-resistant isolates Mycobacterium tuberculosis in Siberia” Dymova M.A., Cherednichenko A.G., Alkhovik O.I., Khrapov E.A., Petrenko T.I., Filipenko M.L. for consideration for possible publication in the BMC section of Infectious Diseases. The manuscript has been read and approved by all authors.

This paper has not been published or accepted for publication. It is not under consideration at another journal.

Sincerely, Maxim Filipenko

The Head of Laboratory of Pharmacogenomics
Institute of Chemical Biology and Fundamental Medicine,
SB RAS, 630090 Novosibirsk, Russia
E-mail: max@nibochnsk.ru;
Telephone number: +7 (383) 3635171;
Fax: +7 (383) 3635171.